Author's response to reviews

Title: Adult brain abscess associated with patent foramen ovale: Case report.

Authors:

Georgios T Stathopoulos (gstathop@med.uoa.gr)
Christina G Mandila (xmandila@yahoo.com)
Georgios V Koukoulitsios (george97@hotmail.com)
Nikodimos G Katsarelis (eglax@hotmail.com)
Michel Pedonomos (eklax@hotmail.com)
Andreas Karabinis (akarab@ath.forthnet.gr)

Version: 2 Date: 10 March 2007

Author's response to reviews:

Dear Editors of the Journal of Medical case Reports,
In response to the Editor's suggestions we:
1. Moved the Competing Interest, Acknowledgements and Author's contributions to the end of the manuscript, after the discussion section.
2. We added a statement to the Acknowledgements list to confirm that patient consent was received for the manuscript to be published.